Author’s response to reviews

Title: Varied overground walking-task practice versus body-weight-supported treadmill training in ambulatory adults within one year of stroke: a randomized controlled trial protocol

Authors:

Vincent G DePaul (depauv@mcmaster.ca)
Laurie R Wishart (wishartl@mcmaster.ca)
Julie Richardson (jrichard@mcmaster.ca)
Timothy D Lee (scapps@mcmaster.ca)
Lehana Thabane (thabanl@mcmaster.ca)

Version: 3 Date: 20 October 2011

Author’s response to reviews: see over
Resubmission of manuscript 1265617550589185: Response to Editors comments.

Thank you for reviewing our manuscript (MS: 1265617550589185) for publication in BMC Neurology. We have revised the manuscript in response to the editor’s comments. Revisions are outlined in detail below.

Editors comments:

1. Affiliations: please provide all of the affiliations on the title page, not just the first one.

   Response: All affiliations were added to the title page.

2. Authors' contributions: Please indicate that all authors read and approved the final manuscript.

   Response: Authors’ contributions revised as instructed.

3. References: all references must be cited within the manuscript text in numerical order. Currently reference 57 is not cited, please cite it in order.

   Response: Reference 57 is cited on page 17, line 13.

References: please provide a functioning link to reference 15

Response: Functioning link added for ref. 15.

Minor revisions

Footer: please remove this.

Response: Footer removed.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Response: Figure 2 revised to reduce white space.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: Manuscript checked for errors and corrections made.